Semester: SS 2021

UnivIS - Lehrveranstaltungsplan

08:00

09:00

10:00

11:00

12:00

13:00

14:00

15:00

16:00

17:00

18:00

19:00

20:00

21:00

Mo

Redlining in America – 1934 to Today (Zehelein)

Redlining in America – 1934 to Today (Zehelein)

Introduction to English and American Literary Studies (A) (Brähler)

American Literature I (Colonial Era - Romanticism) (Gerhardt)

Indigenous America in the 21st Century: A Cultural Examination (Feier)

Indigenous America in the 21st Century: A Cultural Examination (Feier)

Di

Feminist Fiction – Pre and Post #MeToo (Zehelein)

Introduction to English Linguistics (Krug) Online-Meeting

Introduction to English Linguistics

(nur 6.7., 13.7.)

(Krug)

Jews in England: From William the Conqueror to Oliver Cromwell and Beyond (Houswitschka)

How to Write a Term Paper (inkl. Betreuungsübung BA Amerikanistik) (Konopka)

How to Write a Term Paper (inkl. Betreuungsübung BA Amerikanistik) (Konopka)

Mi

New Approaches to Old Favorites: Re-Reading North American Children’s and Young Adult’s Book Classics (1860s-1960s) (Konopka)

Introduction to British and American Cultural Studies (Bickert)

Do

Introduction to English and American Literary Studies (Course B) (Konopka)

Malcolm X, die Nation of Islam und die afroamerikanischen Muslime (Franke)

Fr

"This Double-Consciousness": African American and Afro German Literature and Culture

(nur 23.4., 7.5.)

(Spychala)

"This Double-Consciousness": African American and Afro German Literature and Culture

(nur 23.4., 7.5.)

(Spychala)

Jews in England: From William the Conqueror to Oliver Cromwell and Beyond

(nur 9.7.)

(Houswitschka)

Sa

"This Double-Consciousness": African American and Afro German Literature and Culture

(nur 24.4., 8.5.)

(Spychala)

"This Double-Consciousness": African American and Afro German Literature and Culture

(nur 24.4., 8.5.)

(Spychala)

Jews in England: From William the Conqueror to Oliver Cromwell and Beyond

(nur 10.7.)

(Houswitschka)

So

Jews in England: From William the Conqueror to Oliver Cromwell and Beyond

(nur 11.7.)

(Houswitschka)

Stand: Freitag, 20 September 2024 13:17:18